Author's response to reviews

Title: Early termination of ISRCTN45828668, a phase 1/2 prospective, randomized study of Sulfasalazine for the treatment of progressing malignant gliomas in adults

Authors:

Pierre A Robe (pierre.robe@ulg.ac.be)
Didier H Martin (didier.martin@chu.ulg.ac.be)
Minh T Nguyen Khac (mt.nguyen@student.ulg.ac.be)
Maria Artesi (maria.artesi@gmail.com)
Manuel Deprez (Manuel.Deprez@ulg.ac.be)
Adelin Albert (aalbert@ulg.ac.be)
Sophie Vanbelle (sophie.vanbelle@ulg.ac.be)
Stephane Califice (stephane.califice@hotmail.com)
Markus Bredel (m-bredel@northwestern.edu)
Vincent Bours (vbours@ulg.ac.be)

Version: 5 Date: 1 October 2009

Author's response to reviews:

As requested, the trial registration was added as the final section of the abstract, and the author's contributions have been disclosed.